Author’s response to reviews

Title: The prognostic significance of lymphovascular invasion in patients with resectable gastric cancer: a large retrospective study from southern China

Authors:

Peng Li (lipeng@sysucc.org.cn)
Hao-Qiang He (hehq@sysucc.org.cn)
Chong-Mei Zhu (zhuzhm@sysucc.org.cn)
Yi-Hong Ling (lingyh@sysucc.org.cn)
Wan-Ming Hu (huwm@sysucc.org.cn)
Xin-Ke Zhang (zhangxk@sysucc.org.cn)
Rong-Zhen Luo (luorz@sysucc.org.cn)
Jing-Ping Yun (yunjp@sysucc.org.cn)
Dan Xie (xiedan@sysucc.org.cn)
Yuan-Fang Li (liyuanf@sysucc.org.cn)
Mu-Yan Cai (caimuyan@hotmail.com)

Version: 5
Date: 8 April 2015

Author’s response to reviews: see over
Mr. Ryan Relox and Dr. Wei Jiang,

Editor,
BMC Cancer
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom
E-mail: bmccancer@biomedcentral.com

MS: 4096263541518415 R2

Title: The prognostic significance of lymphovascular invasion in patients with resectable gastric cancer: a large retrospective study from southern China

Dear Mr. Ryan Relox and Dr. Wei Jiang,

Thank you for reviewing the above-referenced manuscript submitted earlier to your office. We would like to take this chance to express our appreciation to you.

In accord with the Editorial request, this manuscript has been thoroughly edited by a native English speaker from an editing company. Editing Certificate will be provided upon request. We feel that the revised manuscript has been strengthened by the Editorial suggestion.

The format for our revised manuscript has been properly prepared and conforms to BMC Cancer style.

If there are any questions or problems for our revised manuscript, please feel free to contact me.

Best regards,

Mu-Yan Cai, M.D., PhD.,
Department of Pathology,
Sun Yat-sen University Cancer Center,
No. 651, Dongfeng Road East, Guangzhou, 510060 China
Tel: 86-20-87342269; Fax: 86-20-87343268
Email: caimuyan@hotmail.com